Author’s response to reviews

Title: HIV-1 Integrase resistance among antiretroviral treatment naive and experienced patients from Northwestern Poland

Authors:

Miłosz Parczewski (mparczewski@yahoo.co.uk)
Magdalena Leszczyszyn-Pynka (mlpynka@interia.pl)
Dorota Bander (dbander@interia.pl)
Anna Urbańska (annaurbanska24@wp.eu)
Anna Boroń-Kaczmarska (annabk@inet.pl)

Version: 2 Date: 8 September 2012

Author’s response to reviews:

Dear Sirs,

the suggested revisions: clarify if samples taken as part of standard care within the manuscript and if informed consent was provided, please also name the ethics committee that approved the study. were implemented into the methods section (lines 89-92).

Thank you very much for these revisions.

with kindest regards,

Milosz Parczewski